Author's response to reviews

Title: Prevalence and determinants of asthma in adult male leather tannery workers in Karachi, Pakistan: A cross sectional study

Authors:

Khurram Shahzad (drkhurrambaig@hotmail.com)
Saeed Akhtar (saeed.akhtar@hsc.edu.kw)
Sadia Mahmud (sadia.mahmud@aku.edu)

Version: 5 Date: 4 December 2006

Author's response to reviews:

December 2, 2006
The editor,
BioMed Central,

Subject: Revised submission of MS: 1231406977103996 - Prevalence and determinants of asthma in adult male leather tannery workers in Karachi, Pakistan: A cross sectional study

Dear Editor,

Thank you for accepting the manuscript. The final formatting changes have now been made.

Thank you and best regards,

Dr Khurram Shahzad
MBBS, MS